Author’s response to reviews

Title: Esophageal squamous cell carcinoma presenting with extensive skin lesions: a case report

Authors:

  Gabriela B. Iwanski (g.iwanski@uke.uni-hamburg.de)
  Andreas Block (block@uke.uni-hamburg.de)
  Gunhild Keller (keller@uke.uni-hamburg.de)
  Julia Muench (muench@uke.uni-hamburg.de)
  Susanne Claus (claus@bakb.net)
  Walter Fiedler (fiedler@uke.uni-hamburg.de)
  Carsten Bokemeyer (c.bokemeyer@uke.uni-hamburg.de)

Version: 9  Date: 9 April 2008

Author’s response to reviews:

Dear BioMed Central Editorial Production Team

Author,

I made all the required changes:

The list of authors in the manuscript is written exactly as they are in the submission system.

Abstract: The abstract is structured into the following sections:

  Introduction
  Case presentation
  Conclusions

Consent: the consent section is now between the authors contributions and acknowledgements.

Thank you for publishing the case report.

With best regards,

Gabriela Iwanski